

Urban Mass Transportation Administration

TRANSIT MARKETING MANAGEMENT MARKET RESEARCH

Prepared for:

U. S. DEPARTMENT OF TRANSPORTATION Urban Mass Transportation Administration Office of Transportation Management Washington, D.C. 20590

ITA Technical Assistance Program

HE 4451 .T83 1988



MTA LIBRARY

1. Report No.	2. Government Acces	sion No. (NTIS)	3. Recipient's Catalog M	lo.		
		(
UMTA-URT-30-88-1						
4. Title and Subtitle			5. Report Date			
Transit Marketing Manao	oment Market R	esearch	March 1988			
Transit Marketing Management Market Research.			6. Performing Organizati	on Code		
		F	8. Performing Organizati	an Report No		
7. Author's) J. Caruso, Edit	or *		5. renorming Organizati			
ou baiabo, lait						
9. Performing Orgonization Name and Addre		10. Work Unit No. (TRAI	S)			
		URT-30-87-1				
Grey Advertisin	Г	11. Contract or Grant No).			
777 Third Avenu						
New York, New Y	ork 10017		13. Type of Report and F	Period Covered		
12. Sponsoring Agency Name and Address						
U.S. Department of Trans	portation		Marketing Re	search		
Urban Mass Transportatio						
400 Seventh Street, S.W.			14. Sponsoring Agency C	ode		
Washington, D.C. 20590			URT-30			
15. Supplementary Notes						
*Edited b	y the American	Public Transit	Association (A)	PTA)		
		e Caruso, Chairn				
Subcommit	tee.	-				
16. Abstract						
This report offers a bro	ad overview of	concumor rocon	rah ag it appli	on to the		
specific problems of tran						
upon data gathered as pa						
in Baltimore, Maryland an						
designed to provide tran						
research, its functions						
in a marketing program.	ine report em	phasizes basic f	narketing princ	ipies rather		
than detailed techniques	; IL IS proble	m-oriented not	echnique-orien	ted. Ine		
purpose of this particul						
useful in making marketi	ng decisions s	o that managers	and researcher	s can work		
together effectively. T						
the role of consumer research in the marketing decisionmaking processwhat to research and when to do it. Part 2 describes the types of strategic research						
used at different stages						
research techniquesthe						
Part 4 discusses professional research serviceshow to select them and work with them. A bibliography of marketing research material is included in this						
report along with 4 tech						
sample size, full questi	omarre, locus	groups and othe	er markering er	ements.		
17. Key Words Transit Marks	ting Cuil-	18. Distribution Statem	ent .			
Transit Marketing Guide						
			ilable to the H	-		
			onal Technical Information Service 5), Springfield, Virginia 22161.			
Guidelines Services (NTIS), Sp Marketing Techniques Problem-Solving			ngileid, Virgir	11a 22161.		
Costs Communications 19. Security Classif. (of this report)	Management 20. Security Cles	sif. (of this page)	21. No. of Pages	22. Price		
Unclassified Unclassified 60						
Form DOT F 1700.7 (8-72)	Reproduction of co	npleted page authorized	4			

.

This report was prepared under contract to the U. S. Department of Transportation, Urban Mass Transportation Administration, Office of Transportation Management. The original work was performed by Grey Advertising, Inc., New York, New York. Final re-write and editing was done by the Office of Statewide Transportation Programs, University of Wisconsin–Milwaukee, under the direction of the Marketing Steering Committee of the American Public Transit Association.

The authors wish to express their appreciation for the assistance of transit systems around the country who offered their advice and reviewed the draft documents.

This document is disseminated under the sponsorship of the Department of Transportation in the interest of information exchange. The United States Government assumes no liability for its content or use thereof.

DISCLAIMER

The U.S. Government does not endorse products or manufacturers. Trade or manufacturers' names appear herein solely because they are considered essential to the object of the report.

This document is disseminated under the sponsorship of the Department of Transportation in the interest of information exchange. The U.S. Government assumes no liability for its contents or use Thereof.

	23488
HE	
4451	
. 783	NOV. 0 7 1997
1988	

Table of Contents

Page

Introduc	troduction						
Part 1:	: The Role of Consumer Research in Transit Marketing						
	 A. Understanding Consumer Research						
Part 2:	Types of Research						
	А. В. С.	B. Pre-Implementation Research					
Part 3:	t 3: Technique						
	А. В.	Qualitative Research1. Focus Group Discussions2. Depth Interviews3. Small Scale SurveysQuantitative Research1. Sampling Plan Development2. Questionnaire Development3. Interviewing Methods and Procedures4. Data Preparation5. Data Analysis and Interpretation	8 8 8 10 12 16 18 19				
Part 4:	Pro	fessional Research Services	21				
	А. В. С.	Guidelines for Selecting a Market Research Supplier	21 23 24				
Bibliogr	aphy	y of Marketing Research Textbooks	26				
Technical Appendices							
	А. В. С. D.	Guidelines for Determining Sample Size Strengths and Weaknesses of Interviewing Methods Focus Groups Survey of Surveys					
Exhibits	5						

- A. Sample Screener Questionnaire
- B. Full Questionnaire
- C. Examples of Secondary Information Sources
- D. Comparative Assessment of Alternative Survey Methods

Introduction

This report offers a broad overview of consumer research as it applies to the specific problems of marketers of transit. Some of the information included is based upon data gathered as part of a Transit Marketing Demonstration Project conducted in Baltimore, Maryland and Nashville, Tennessee by the Urban Mass Transportation Administration. The purpose of this project was simple—to demonstrate to the transit industry that modern marketing techniques could be used to increase ridership on public transportation. Implementing this project involved several elements of marketing which have long been applied to a broad array of consumer products and services. One of these key elements which had become standard in other fields, but was relatively new to transit, was market research. Grey Advertising, Inc. was commissioned by the Urban Mass Transportation Administration to conduct this research. Reports on their research findings and other aspects of the demonstrations have been issued under separate cover. (See the Transportation Research Board's *Special Report 184, Urban Transport Services Innovations,* "Service Planning and Marketing, pp. 79-84.)

Since that time, there have been several other UMTA sponsored reports on marketing, and there has been a unified theme to their conclusions: that marketing research in transit should be "customer oriented," that is, it should delineate market segments for target marketing, solicit and interpret a wide range of attitudinal data from current and potential customers, and investigate the relation between stated intentions and later behavior of these individuals. In addition, it has been urged that transit agencies standardize and coordinate their data gathering instruments and procedures, so that more accurate comparisons can be made between comparable markets and market segments.

This document is intended to give transit marketers the tools to accomplish these goals by providing them with a better understanding of consumer research, its functions and limitations, so that they can use research effectively in a marketing program. It is not intended to make the reader a research expert any more than a single chapter on transit marketing could convert a researcher into a marketing expert. Therefore, this report will concentrate more on basic principles than detailed techniques. Consumer research is a more valuable tool when it is *problem*-oriented rather than *technique*-oriented.

The purpose of this particular report will be to describe the kind of research that is most useful in making marketing decisions so that the transit manager and the marketing researcher can work effectively together. The transit marketer must communicate the nature of the transit system and the issues he or she would like to explore to the marketing researcher. The researcher should be able to work with the transit marketer to decide which research techniques would be most appropriate. Through a close, team-oriented approach, the two can pool their expertise to get meaningful research data to solve real-world problems.

The report is divided into four parts:

Part 1 outlines the role of consumer research in the marketing decision-making process—what to research and when to do it.

Part 2 describes the types of research used at the different stages in the marketing program.

Part 3 discusses some of the research techniques—the tools and methods used in the different types of studies.

Part 4 discusses professional research services—how to select them and work with them.

A bibliography of marketing research material is attached to provide the interested reader with greater detail than is possible to cover within the scope of this document.

Part 1: The Role of Consumer Research in Transit Marketing

A. Understanding Consumer Research

Consumer research is a versatile marketing tool with a multitude of uses. In addition to commercial applications, it can be used to pursue political and humanistic goals. For example, consumer research can deal with merchandise such as canned soup, services such as public transportation, or political issues such as tax cuts.

Marketing strategies, whether applied to soup or transportation, are based on satisfying the needs of the buyer/user. In the context of public transportation, this means responsiveness to the service and information needs of *current* and *potential* riders. If transit marketing is to be effective, service offerings (headways, routing, safety programs, etc.) and communication/information efforts (advertisements, maps, schedules, etc.) must meet consumer needs.

Market research is a way to identify these consumer needs by gathering data regarding:

- Size and nature of the transit market, both current and potential.
- Consumer awareness of the transit system or a specific aspect of that system.
- Consumer attitudes toward the transit system or a specific aspect of that system.
- Competitive framework within which the transit system operates (e.g. car vs. bus) or specific parts within that system (e.g. subway vs. bus).
- Benefits current and potential riders of system perceive they could gain.
- Barriers to using the system or using it more often. (Do consumers believe that bus travel would take too long, or are they afraid of being mugged at the bus stop?)

The information gathered is used for several purposes:

- to help develop the marketing program;
- to guide the implementation of the program; and
- to refer back to when monitoring program effectiveness.

B. The Marketing Process

Consumer research plays an important role in every stage of the marketing process:

STAGE 1: The marketing process starts with *strategic research* which supplies information for developing the marketing strategy—e.g., who are the current riders, who are the potential riders and why they don't make greater use of the facilities, how can the transit operator better service the public, etc. *STAGE 2:* "Product" or service improvements are planned and an information program is developed to inform target consumers of new transit services and facilities and to persuade them to use the services. This is the time for *pre-implementation evaluation research* which supplies consumer reaction to crucial parts of the program before their implementation at high cost.

STAGE 3: Once the marketing program is implemented, it needs to be monitored to track its effectiveness and determine where adjustments may be necessary. This is the time for *market response analysis* and *tracking and assessment*.

Part 2: Types of Research

A. Strategic Research

Definition

The purpose of strategic research is to gather appropriate data about consumers in the transit operating area and potential operating area so that an effective marketing strategy for the transit system can be developed. On its most general level, strategic research explores consumers' travel behavior, attitudes toward public and private transportation and transportation needs. Specific objectives of strategic research include:

- * Defining the consumer groups offering the greatest potential for increased transit ridership.
- * Identifying the motivators that will encourage these consumers to use transit services. It does this by providing an integrated understanding of their transportation needs, attitudes relating to transportation, and behavior patterns vis-a-vis current public and private facilities.

In order to achieve these objectives, market or consumer research seeks to answer the following questions:

Who is the target customer?

Who currently rides mass transit? Who might switch from private to public transportation? What proportion of the increased trip volume would come from current riders vs. switchers? Which group would be the best target? What is the size of this group in terms of both people and number of additional transit trips? What is the competitive framework? What modes of transportation does public transit compete against by trip type, e.g., business, shopping, leisure, etc.? For which of these trip types is it possible to attract new business away from competitive modes? In addition, what proportion of people would take particular trip types or use a particular service if these became available to them? How many trips would this last group of people account for?

Are changes in the system necessary to tap consumer potential?

When: Any seasonal patterns of usage? Is ridership different during the school year compared to summer? At what time and on what days, e.g. rush hour Monday-Friday?

Where: From what points to what destinations and for which purposes do the best transit prospects travel within the system's operating area?

Pricing: What are the benefits of raising/lowering fares, transfers, special fares? How would such fares affect trip volume? Revenue? What motivating appeals are necessary to overcome ridership barriers?

Which benefits do people seek from the transport modes they use? Why do they use transit? What problems should be overcome to stimulate greater use of transit?

What should be stressed in communicating with customers?

Which benefits should be emphasized to motivate ridership? How should the target consumers be addressed? What tone should be used to appeal to their lifestyle? Which media will provide the maximum impact?

Types of Strategic Market Research

There are two main types of strategic research studies:

(1) MARKET WIDE STRATEGIC STUDY: When the research is concerned with an entire transit system within an entire metropolitan area.

(2) LIMITED AREA INCENTIVE STUDY: When the research is concerned with a specific route, neighborhood, or with a single system service.

Market-Wide Strategic Research

Although market-wide strategic research is expensive and time-consuming, such a study can provide a thorough understanding of consumer needs and desires. Opinions differ on how often market-wide strategic research studies should be conducted. Certainly if the demographics of your geographic area are changing, or if your market is altering for any other reason, such a study should be done every 3 to 5 years. It will supply the key information outlined above and identify:

- * Target market.
- * Attitudes, transportation needs, information needs, travel behavior, basic service satisfaction and dissatisfaction of current and potential riders in the operating area.
- * The benefits which the communications program should stress and the tone the communications should adopt to achieve maximum impact.

The analysis of data collected from a market-wide strategic study lays the groundwork for a marketing plan for the entire transit area. However, before deciding to conduct a market-wide strategic study, it is suggested that the marketing manager examine studies conducted in other metropolitan areas, particularly similar areas. He or she may find that much of the information acquired in another area applies to his or her area. In this case, a smaller piece of research could be conducted to validate the applicability of data collected elsewhere to local conditions, thus saving time and money. Looking at studies from similar areas can also give you an idea of what types of questions yield the most useful information. Even if the demographic characteristics are different, it is still possible that the research design and methodology will be relevant.

Limited Area Incentive Studies

A limited area incentive study is a modified form of strategic research designed to gather relevant information regarding:

- Need for and/or market potential of a specific single change in the transit system, or
- * One key market segment or mass transit need in one neighborhood.

As noted earlier, a limited area incentive study can be used to develop or modify a single component of a marketing plan or strategy. Since the specific issues addressed in a limited area incentive study vary, the nature of each study should be specially designed to best address the issue in question.

Some examples of topics for limited area incentive research are:

- * a routing study to determine if the routes serving a particular neighborhood satisfy consumer origin and destination needs.
- * a rider definition study to identify and locate groups of potential riders for a specific route or corridor, and the types of service improvements required to induce greater use of that route.
- * a rider incentive study to determine consumers' image of transit service so that misconceptions about routes and special services can be addressed through better communications programs and information aids.
- * an evaluation of a specific service such as park and ride.
- * a needs assessment of a specific group such as senior citizens, handicapped riders, etc.

Limited area incentive studies should be conducted before implementing a service to determine whether there is truly a need for such a service from the consumer point of view, and what the market potential of such a service would be. In addition, when a service has been implemented, but has failed to meet its projected market potential, a limited area incentive study can answer some vital questions about why a service is *not* reaching projected market potential. By checking the original projections, comparing them to actual after-the-fact data, and running a second study, it is possible to learn what action is necessary to assure that the service will reach market potential.

Since the scope of a limited area incentive study is generally narrower than the scope of a market-wide strategic study, the latter generally has less stringent sampling requirements and costs less to conduct.

B. Pre-Implementation Research

An analysis of the data generated by market-wide strategic research and limited area incentive research can provide certain necessary information: the need for service improvements, the design of communication programs and information systems, etc. However, since a major effort and expense is required to implement service changes and advertising campaigns, it is frequently worthwhile to get consumer feedback before implementing any part of a new plan. This preservice evaluation will help determine if the plan will accomplish its objectives and if the money will be spent effectively.

Of the three basic types of consumer research discussed in this report, pre-implementation research will probably be the most frequently used by transit marketing managers. When questions arise regarding consumers' reactions, acceptance, or comprehension of a new product or service, pre-implementation evaluation is a means of obtaining consumer feedback relatively quickly. This kind of study is usually tailored very specifically to the situation at hand; that is, the objectives, design, scope and method of individual pre-implementation studies are determined by the specific problem to be addressed, making each such study a custom study. Consequently, the problems that may be explored in such research are as extensive and varied as the marketing program itself. However, they may be categorized in three general areas:

Concept tests:

Used to evaluate new ideas on the "drawing board," including new services, changes in service, or possible new products, such as timetables or bus shelters. The new ideas or concepts are presented to consumers in visual and/or verbal form, as appropriate to the concept being evaluated. Consumers are then asked about their reactions to the overall concept, attitudes on specific benefits/problems associated with the concept, likes and dislikes, likelihood to use, etc. Since concepts are frequently hypothetical, and often refined after applying information gathered from a concept test, marketers should always consider following up concept tests with "product" tests, when the product permits such evaluation.

Product tests:

Used to evaluate a prototype of an actual product such as route maps, timetables, or bus shelters or to evaluate a particular service such as "Park and Ride" lots or "Shoppers Specials." Consumers examine the product or try a test run of the services. They are then asked about their reactions to the product or service in terms of attitude overall, attitudes on specific benefits/problems, needs satisfaction, likes and dislikes, etc. (Obviously, product tests are not feasible for some services; for example, a new subway system). By determining if the new product or service will fill an unsatisfied need, and the relative market potential of such a service in terms of trip volume, the marketer can determine whether to proceed with the proposed change or to cancel or modify it.

Copy tests:

Deal exclusively with evaluation of an advertising execution. Copy tests expose current and potential riders to an execution to determine if it:

- * Does not cause any negative reaction.
- * Communicates the intended message—clearly, concisely, convincingly.
- * Attracts attention.
- * Persuades the viewer of the key benefits of the service being advertised.

* Persuades the viewer to use the service being advertised.

By correctly analyzing the information gathered in a copy test, the marketing manager has the opportunity to improve an execution before it runs. As a result, the marketer will know that advertising dollars are working as effectively as possible.

The same conceptual approach can be used to evaluate whether other communication/information aids, such as brochures, timetables or route maps, are effective.

C. Tracking and Assessment

Once a new marketing plan, or a major portion of a plan, is implemented, it is essential to learn whether the plan is accomplishing its objectives. Market response analysis, such as passenger counts and revenue counts, evaluate the business or financial side of a program. However, such analysis does not, and cannot, explain why business is improving, declining, or remaining level. This kind of evaluation requires tracking or assessment research. Tracking research involves repeated sampling of a specified population, asking the same questions each time or a core of standard questions, but with a new selection of respondents.

Tracking, if designed properly, will provide a marketer with information regarding *why* business is up, down, or the same from a *consumer perspective*. Tracking usually involves a series of two or more identical surveys conducted over time. The first survey or wave of the series is commonly called the "benchmark" wave. This wave is conducted just prior to implementing a service or product being tracked. This wave provides the benchmark levels of:

- Consumer usage or ridership.
- Consumer awareness of service.
- Consumer attitudes toward service.

Follow-up surveys or waves are conducted at various intervals after the new service or product has been implemented. Results of successive waves are compared to earlier waves and the benchmark wave in order to determine the degree and direction of change over time on all of the above measures. For example, if a transit marketing manager learns why ridership on a new route is down, he may implement corrective action. The next wave of the tracking research should tell the marketer if the corrective action was effective.

The number and timing of subsequent "waves" depends on the implementation schedule for the marketing program. When a service improvement and communications package are initiated at the same time, two subsequent survey waves are recommended. The first should be conducted shortly after the implementation of the package, but with sufficient time for the marketing program to have had a measurable effect. For example, if the communication package involves a 4 week advertising campaign, the first wave of the research should be conducted within a few days of the end of the campaign. The next wave should be conducted after the program has been in operation enough time for it to have full effect (possibly a year). When a marketing plan is implemented in several stages, additional follow-up waves are desirable. Their timing, however, is crucial.

Part 3: Technique

The purpose of this section is to highlight various methodological issues that frequently arise when conducting consumer research. Before going further, though, it's necessary to understand the important distinctions between qualitative and quantitative research. These terms are frequently misunderstood, even by knowledgeable marketers.

A. Qualitative Research

Qualitative research is a valuable but limited marketing research tool. It can be a simple and inexpensive way to get good feedback from current and potential riders about such things as advertising copy, schedules, maps, graphics, jingles, and other elements of the transit service. However, it's important to remember that qualitative findings are not projectable to the larger population. Therefore, it is crucial that respondents be recruited in such a way as to accurately reflect, as closely as possible, the actual audience for a product or service. One way of putting it is that the purpose of qualitative research is to add meaningful input to quantitative research. For example, one can learn what language or words consumers use to describe a particular transit-related product, service or issue. This language can then be used in framing the questions for a quantitative study. Qualitative research may also be used to generate hypotheses, stimulate ideas, and develop background information about consumer attitudes, beliefs and predispositions.

Some examples of qualitative research are:

- Focus group discussions—a talk session among 10-14 people under direction of a "moderator" who explores in-depth an issue using a guide prepared ahead of time by a team composed of the marketing manager, the researcher, and the person moderating or facilitating the group. (See Technical Appendix C: Focus Group Discussions.)
- 2. Depth interviews—a free-hand one-to-one talk session between a single consumer and an interviewer who probes the depths of the person's knowledge and/or opinions on a specific subject. This technique consists of probing more deeply into initial responses to questions.
- 3. *Small scale surveys*—a semi-structured interview with 10-25 individuals in which an interviewer asks a series of structured (pre-established) questions and probes responses in limited depth on subjects pertinent to the issue.

Qualitative research by its very nature is not decision-oriented and cannot be used to solve marketing problems. Its sampling universe is restricted, the information derived is scattered and non-objective; it represents the viewpoint/experience of a very limited number. It is important that the use of qualitative research be carefully controlled by the transit marketer to insure that the technique is not misused.

B. Quantitative Research

Quantitative research, on the other hand, usually relies on a much greater sample and is the basis for decision-making. It lays the ground for a study of broad scope, carefully designed and

properly sampled. Quantitative research is sometimes divided into applied research and descriptive research.

Applied research: Used when a problem or issue has already been tentatively identified and management wants to get data that would help solve or illuminate it. Typical applied transit research projects would be to find causes of declining ridership on a specific route, to determine the reasons for poor morale among drivers, or to evaluate advertising alternatives.

Descriptive research: Used to gather background data which helps management better understand their service and users. Transit agencies may be interested in the characteristics of riders and non-riders; they may want data about the behavior and attitudes of people within their service area. Another possibility is finding out more about the trip destinations and characteristics of their current and potential riders.

Quantitative Research Methodologies

There is no single way to conduct consumer research, no single technique that is appropriate for every type of marketing problem, and no single methodology that will provide valid results in all research situations. The specific methodology to be used in a study depends upon the definition of the marketing problem and the conceptual approach to it. For a very useful compilation of current mass transit survey questionnaires, see *Use of Market Research in Public Transit* (1985) prepared by the Center for Urban Transportation Studies, University of Wisconsin–Milwaukee.

Since there is such a close relation between the nature of the problem or issue in question and the technique used to study it, it is very important that the transit marketing manager and the researcher each have a clear understanding of the objectives of the study. In order to set specific goals, the problem or issue must first be clearly defined. Only then is it possible to set reasonable objectives by asking such questions as: What exactly is the study going to accomplish? What kind of "actionable" information will be gained? What segment of the market will be targeted for responses? What are the limits of the research?

Once the research problem has been defined and the objectives agreed on, the researcher and marketer can begin structuring their study. Before developing or selecting a methodology, they must consider:

- Nature of the problem and degree of risk in various solutions.
- Extent of data requirements.
- Nature of measurements to be made.
- Time and cost constraints.

Valid, reliable, and useful consumer research provides information on relevant issues which in turn enables management to make informed decisions. For this reason, the design and execution of quantitative studies should adhere to basic research principles in the following areas.

- 1. Sampling plan development.
- 2. Questionnaire development.
- 3. Interviewing methods and procedures.
- 4. Data preparation.
- 5. Data analysis and interpretation.

Note: No research problem is unique. Generally some information already exists that will aid in refining the questionnaire. Such sources of information include previous research conducted by the local transit authority, planning agencies, state or Federal transportation departments or other agencies such as the Census Bureau, as well as local colleges and universities. (See Exhibit C: Examples of Secondary Information Sources.)

Additional background information may be desirable to formulate hypotheses, develop phrase lists for attitude measurements, or learn what language consumers use to discuss the subject. For this type of preliminary exploration, qualitative research is used (group sessions, depth interviews, etc.)

1. Sampling Plan Development

After defining research objectives and selecting a methodology, the next task involved is developing a sampling plan. A sample is a finite part of a statistical population whose characteristics are to be studied in order to gain information about the whole population.

Developing a sampling plan consists of determining the:

- * universe or "population" to be included in the study;
- * type of sample to be drawn;
- * size of the total sample and the necessary sub-groups to be analyzed.

The Sampling Universe

A sampling universe is the entire relevant population from which a sample will be drawn. Generally, a sampling universe for public transportation studies has two dimensions:

(1) Geography and (2) Type(s) of people.

The *geographic* universe for a sample is determined by the scope and intent of the study, and could involve an entire transit area, one neighborhood, or the area within a certain distance of a route. For example, the geographic boundaries of a systemwide study would encompass the entire transit operating area. For a single route study, the geographic boundaries would consist only of those areas served by the route. The characteristics of the geographic territory determine the precise description of the geographic boundaries.

These geographic characteristics will vary due to:

- Nature of the geographic constraints, i.e., the physical layout of the service territory.
- Density of the transit network.
- Presence or lack of multiple transit modes.
- Extent to which consumers in suburban or outlying areas have an opportunity to use existing or feasible new transit services, e.g., "park and ride" or commuter rail-bus lines.
- Franchise or other restrictions.

The type of people who make up a sampling universe is also determined by the scope and intent of the study, and could involve the entire population in a transit area or only one segment of the population, such as senior citizens, those living within a certain distance of a route, etc. For example, the universe for a market-wide strategic sample would encompass the entire population in the transportation area. If evaluating the potential of a special fare for senior citizens the sample universe should be composed of senior citizens; for evaluating the potential of a special rush-hour route, the sampling universe would encompass those who live/work within a fixed distance of the route and travel at rush-hour.

Type of Sample

Sampling can be highly complex and technical. It is suggested that the transit marketer rely upon a consultant or research supplier to help determine the appropriate sampling universe and/ or draw the sample. Our discussion will highlight the nature of:

- * Full probability sampling.
- * Modified probability sampling.
- * Quota sampling.

(For further details on sampling, see Technical Appendix A or refer to the statistics and market research sources in the Bibliography.)

Full Probability Sampling

This type of sample adheres strictly to all rules of sampling so that each member of the defined universe has both a *known* and *equal* chance of selection. In addition, all field controls (e.g., callbacks, respondent selection procedures, validation, etc.) are followed rigorously in order to ensure that the sampling procedure is properly administered. Such efforts are needed to successfully avoid respondent selection bias by the interviewer or the problem of the availability of some people and not others.

Data collected from full probability samples are fully projectable to the entire target population. However, since full probability samples are generally quite expensive, they are used only when absolutely necessary, that is, when the decisions being made involve high risks. For example, one would probably use a full probability sample for a major market-wide strategic study or for a study requiring accurate volume projections. Otherwise, less stringent sampling can be used.

Modified Probability Sampling

When the decision being made is of a moderate risk level, a completely representative and fully projectable sample is often not necessary. In this case, the transit marketer can reduce research costs by using a modified probability sample. In such a sample, some of the rigid rules and field procedures of a full probability sample are sacrificed. For example, one might select fewer sampling points or locations. Since fewer sampling points increase the efficiency of the interviewer's time, savings can result from reducing points below that required for fully projectable results. However, respondents must still be selected on a random basis from those sampling points used, if the results are to be at all representative.

Quota Sampling

When the decision being made involves relatively low risk, a quota sample may suffice, particularly if the universe consists of one or two specific segments of the population. Quota sampling requires interviewers to reach out and find eligible respondents as they come across them, until the number of interviews reaches the pre-designated sample size. However, in order to guarantee that the relevant population is included in the sample, specific eligibility requirements are established. Potential respondents are screened to see if they meet these eligibility requirements before a full length interview is conducted. This method is often used for product tests.

Determination of Sample Size

Sample size must always be determined in advance, regardless of the type of sample one is using. This is a very important task. Sample size is partly based upon the number and nature of the sub-groups being analyzed. For example, in strategic studies, it is necessary to have a large enough total sample size to permit valid analyses of various sub-groups such as heavy riders, moderate riders, light riders, and non-riders. Each sub-group should have a minimum of 100 respondents.

When conducting pre-implementation evaluation research and tracking or longitudinal research, it is necessary to first decide how much statistical sampling error will be tolerated. Then the appropriate sample size can be calculated, one that will permit you to stay within the desired range of sampling error. (Any statistics textbook will have formulas for this process.)

2. Questionnaire Development

The purpose of a questionnaire is to collect data in an organized and objective manner. It is a tool used to gather relevant information from consumers which will later be analyzed. Appropriate marketing decisions can then follow analysis. The length and form of a questionnaire will vary according to the objectives and budget of the study. It is necessary to develop a different set of questions for each research project, with data tailored to meet the objectives of each specific study. A good questionnaire asks the right questions in proper sequence, in clear, unambiguous, unbiased language. Long questionnaires should also be constructed in a way that avoids respondent fatigue. For example, if there are a lot of rating items, they should be broken up into two or more questions to encourage respondents to give a more thoughtful answer. In personal interviews, various props such as cards with answer choices can be used to alleviate respondent fatigue.

Questionnaire development is a critical step in the research process because the usefulness of the data collected can be no better than that of the questionnaire itself. Therefore, it is recommended that transit marketers rely on someone with extensive experience in questionnaire writing.

Designing the Questionnaire

The development of the actual questionnaire involves a team effort by a transit marketer and a researcher.

The MARKETER'S ROLE is to provide a clear statement of the problem, the marketing issues, and objectives of the project.

The RESEARCHER'S ROLE is to translate the above information into a questionnaire that will accomplish the objectives of the study.

The procedure is one of give and take. The marketing needs are assigned priorities to ensure proper focus of the questionnaire. One of the greatest difficulties in drafting a questionnaire is

resisting the temptation to ask everything. The content must be confined to the most relevant issues.

Questionnaire Content and Techniques

Although the content of questionnaires varies widely, there are two basic questioning techniques: STRUCTURED CLOSED-END questions and UNSTRUCTURED OPEN-END questions.

STRUCTURED CLOSED-END questions limit respondents' answers to predetermined possibilities such as yes/no, true/false, rating scales, and other multiple choice questions. The respondent merely selects his or her answer from a list.

Rating scales are a widely used form of structured closed-end questions, especially for attitudinal information. Rating or intensity scales use between three and eleven point scales to measure opinions. Deciding how many points to use on a scale depends on general considerations such as the purpose of the survey and the nature of the issue. It can vary between three and eleven, but the most common scales fall in the three to seven range. That is because, generally speaking, discrimination levels off at seven points, with no refinement at all possible after eleven. The Likert scale is a popular example of this type of ranking. It presents five or six degrees of possible agreement:

Disagree	Disagree	Disagree	Agree	Agree	Agree
Strongly	Moderately	Mildly	Mildly	Moderately	Strongly

In a face-to-face interview, the interviewer often gives the respondent a card with the scales printed on it because the range of responses is large. Rating questions, no matter what their point scales, usually elicit the following kinds of responses:

OPINIONS:

excellent, very good, good, fair, not so good, poor.

USAGE INTENTION:

definitely would ride the bus, extremely likely, very likely, somewhat likely, not very likely, not at all likely to ride the bus.

DESIRABILITY OF SPECIFIC BENEFITS of a product or service: extremely desirable, very, quite, rather, slightly, not at all desirable.

Attitude or ranking scales are very sophisticated tools and should be used carefully by those fully familiar with their assumptions and limitations.

In contrast to the highly-structured questions above, UNSTRUCTURED OPEN-END questions allow the respondent unrestricted expression of opinion; thus they permit more detailed responses than is possible to obtain from closed-end questions. However, open-end questions are somewhat subjective because the response to such questions depends upon the interviewer's ability to effectively probe the respondent, the respondent's ability to verbalize thoughts, and the coder's ability to correctly code the answers (place the responses into categories so that they can be fed into the computer). Therefore, the use of open-end questions should be restricted to questions for which a closed-end answer would not be adquate to elicit the desired information. It is important to understand that proper questionnaire design is a critical component of the survey process and can determine the success or failure of the study. The information desired must be carefully thought through in advance. Factors that must be considered when designing a questionnaire are:

- GET ALL THE INFORMATION NEEDED, but restrain the impulse to ask everything.
- * WORD QUESTIONS ACCURATELY
 - Keep wording as simple as possible.
 - Make sure questions can be understood by all.
 - Avoid slang and technical terms.
 - Do not bias responses.
- * ARRANGE QUESTIONS CAREFULLY
 - The first question should arouse interest.
 - The order of questions should be checked to make sure that one question will not bias the response of another.
 - Break-up monotonous questions; for example, it too many rating items are asked at one time, the respondents are likely to get bored and give a less than thoughtful answer.
 - The order of rating items should be rotated. In other words, one phrase should not always be in the same position on the line (first, second, last, etc.). In this way, no one phrase will be subjected to interest bias because of positioning.
 - Ask all recall or unaided awareness questions first, before respondent can guess purpose of study. These questions and their proper interpretation can provide insightful diagnostics at a later time.
 - Personal questions should be worded in a way that will avoid embarrassing or offending respondents.
- * MAKE QUESTIONNAIRE MECHANICALLY EFFICIENT
 - Make sure there is adequate space for answers.
 - Make sure that question sequence and skip patterns are easy to follow by means of organization, layout and instructions.
 - Transition phrases between question areas should prepare respondents for the next topic area, as well as stimulate interest.

Question Sequence

The sequence in which questions are asked within the survey is crucial to successful questionnaire design. This is still another phase that requires close teamwork between the transit marketing manager and the researcher.

Generally speaking, transit survey questionnaires consist of questions in the following four categories:

- 1. Introduction.
- 2. Warm-up.
- 3. Main study.
- 4. Demographics.

The most difficult questions to ask are those regarding personal characteristics such as age, sex, race, income, religion, etc. That is why the demographic questions should come at the end of the questionnaire, after the interviewer has had the chance to establish rapport with the respondent. Placed earlier in the questionnaire, such personal questions could scare the respondent off or create tension that would affect responses. The designers of the survey need to ask themselves if the information elicited in each question is really necessary. For example, income data by itself is not very useful; it may be more meaningful to determine automobile ownership, a figure that relates more directly to transit use and also can be used as an indicator of income.

Question Areas

Each questionnaire should begin with an introduction which informs the potential respondent of the:

- 1. Name of the interviewer's research organization.
- 2. Purpose of the study.
- 3. Importance of respondent's cooperation.

Following are five general categories of mass transit research questions:

- 1. AWARENESS: How much the public knows about the transit system, including services available, existing routes, stops/terminals, sources of information, fares and advertising.
- 2. CONSUMER BEHAVIOR: What consumers' usage habits are relative to transportation, including frequency of travel, purpose of travel, weekday/weekend usage, time of travel, purpose and use of public transportation facilities and purpose and use of alternative modes of travel.
- 3. CONSUMER ATTITUDES: How consumers feel about a transit service and about competitive modes on the overall and on specific features (such as dependable service, ease of use, economy, comfort, etc.). How satisfied/dissatisfied consumers are with existing services. Would proposed new services fulfill their needs, etc.?
- 4. SERVICE REQUIREMENTS: Whether current routes serve consumer requirements and what changes are desirable, including where riders come from/go to (origin-destination), time of trip, length of trip and amenities.
- 5. CLASSIFICATION OF TARGET GROUPS: Which consumers offer the greatest volume or potential volume of business, their demographic description (age, sex, occupation, education, income), and their life-style characteristics (homebodies, socially-conscious, family-dedicated, footloose, etc.).

Pre-test Questionnaire

Once a questionnaire is drafted, it is pre-tested. Pre-testing is essentially a "dress rehearsal" before the actual fieldwork for the study begins. Pre-testing is a means of checking the structure of the questionnaire and interview length. Pre-testing can also highlight sensitive questioning areas in which respondents may be reluctant to participate; it can also point out questions that are difficult for the respondent to understand or answer. Pre-testing allows for appropriate changes in

the questionnaire before the final version is printed. A standard rule in the field of market research is that the pre-test should be conducted with ten percent of the planned sample size. In cases where portions of the questionnaire are re-designed after a pre-test, a second pre-test should be conducted.

Note: As part of the pre-test, answers should be coded, recorded and analyzed to check out the entire procedure.

3. Interviewing Methods and Procedures

INTERVIEWING METHODS: Interviews can be conducted in person, on the telephone or through the mail. The interview method selected depends upon:

- Type of sample: Full probability. Modified probability. Quota, etc.
- Nature of Interview: Length and complexity of questionnaire. Need to show product or other visual stimuli. Time and budget limitations.

There are four major types of interviews:

- 1. Personal Interviews.
- 2. Telephone Interviews.
- 3. Mail Interviews.
- 4. Combined methods.

(See Exhibit D: Comparative Assessment of Alternative Survey Methods.

Personal Interviews

Personal interviews can be conducted in three different ways: in-home, on-site (such as on a bus or at the bus stop) or in a central location such as a shopping mall.

When personal in-home interviews are conducted, the interviewer goes door-to-door, selecting the appropriate household and selecting the appropriate respondent in each according to prescribed procedures. (These procedures are part of the sampling plan.) Finally, the respondent is interviewed in his or her home.

When on-site surveys are conducted, interviewers screen for eligible respondents at the site, and then personally interview them on the spot. In some situations, the interviewer will give the respondent a self-administered questionnaire to fill out.

When central location interviews are conducted, interviewers go to high pedestrian traffic areas, such as shopping malls, and screen passers-by. Eligible respondents are then taken into an enclosed area for a full-length interview. Central location interviewing is often used when a quota sample is sufficient and/or if the respondents are required to handle a product or view a commercial.

Telephone Interviews

Telephone interviews are a widely-used, cost-efficient method of data collection. The interviewer dials the appropriate telephone numbers and asks for the appropriate person to talk with in each household according to prescribed procedures. (See Technical Appendix B, Strengths and Weaknesses of Interviewing Methods for telephone sampling procedures.) The amount of interviewer time required is considerably less for telephone interviews than for personal interviews, making field work faster and less expensive.

Ideally, telephone interviews should be restricted to one-half hour for maximum quality of response, but there are cases where longer telephone interviews have been successful.

(Note: In most markets, the proportion of households with telephones is sufficient. However, it should be pointed out that in those areas where a large proportion of households [20%] do not have telephones, another interviewing method should be used.)

Mail Interviews

Mail interviews are generally conducted by research companies which maintain a panel of households which participate in surveys on a regular basis. For transit studies of a selected area, the transit marketer would have to select a custom sample from a reverse telephone directory in order to canvass the area affected by a new or expanded route or service. Mail surveys of this type have a significantly lower completion rate than either personal or telephone surveys, and also require longer overall time to complete due to the time lapse between mail-out and return.

Combining Interviewing Methods

For some situations it may be appropriate to combine interview methods. For example, if one is looking for a small segment of the population, such as users of special transportation for the handicapped, one might want to screen them by telephone, make an appointment, and conduct the full-length interview in person in their homes.

INTERVIEWING PROCEDURES: Interviewing is a special skill that requires extensive training so that interviewers can learn to:

- * establish rapport with respondents quickly;
- not influence responses;
- * follow questionnaire procedures.

Without such training, interviewers tend to bias the responses and make many other serious mistakes which can jeopardize the results of the study. Therefore, it is essential to use professionally-trained interviewers, maintain interviewing quality, and follow all basic sampling and questionnaire procedures. Marketing research companies either have their own interviewers or will contract with interviewing services.

Written Instructions

Written instructions should be provided which describe study objectives, sampling procedures, respondent requirements, timetable for field work, and any necessary special instructions such as how to use visual aids.

Interviewer Briefings

Interviewer briefings should be conducted in-person by an experienced supervisor. When a lengthy or complex interview is involved, it is advisable for someone from the market research supplier's central staff and/or the transit agency's staff attend the briefing to ensure that difficult sections of the questionnaire will be executed correctly and that any problems will be resolved to the transit agency's satisfaction. During the briefing, all interviewers should be thoroughly informed about the purpose of the study in order to increase their interest in the assignment. Sampling requirements, call-back instructions and interviewer instructions about the questionnaire should be clearly explained, so that no confusion exists about the proper selection of respondents or administration of the survey. The interview should be reviewed with all interviewers on a question-by-question basis. This procedure is most efficiently accomplished in one group meeting so that instructions are uniform and individual interviewers can hear the questions and concerns of others. Where feasible, it is wise to distribute questionnaires to interviewers in advance of the briefing sessions. In addition, all interviewers must conduct at least one practice interview. This step is critical, particularly with complex questionnaires.

Monitoring

In the early stages of data collection, interviews should be monitored to ensure that the interviewers are administering the questionnaires properly. Questions should be read exactly as written and the interviewer's own opinion should never be discernible, either by word or tone of voice. Auditing three or four interviews per interviewer usually suffices.

Validation

In order to check that the proper respondents were interviewed, a minimum of 10% of each interviewer's respondents should be recontacted by a third party and asked if certain question areas should be covered, and if the interviewer was courteous. This is only one method of evaluating field work quality, but is a vital control which virtually every reputable research company exercises.

4. Data Preparation

After field work is completed, the data must be prepared for analysis, a procedure that requires extensive quality control.

Editing and Cleaning

Editing is the first step in processing. Editing requires that each questionnaire be checked for:

ADHERENCE TO SAMPLING INSTRUCTIONS: If the interview was not made with the proper respondent, it is rejected.

LEGIBILITY: If handwriting is illegible, editors first attempt to decipher responses. When deciphering is impossible, the interviewer is contacted for clarification. If necessary, the question under consideration is excluded from final tabulation.

COMPLETENESS: Editors review any blanks to determine whether all questions have been answered.

CONSISTENCY: Each questionnaire is reviewed for discrepancies (for instance "brand of car owned" recorded under one question and "do not own car" recorded in another). The editor also confirms that the skip patterns have been followed accurately (e.g. if "no" in Q. 3, skip to Q. 6). Questionnaires are thus "cleaned" to insure that erroneous data are not recorded.

UNDERSTANDABILITY: Open-end responses are often difficult to decipher because of abbreviations used by interviewers while recording verbatims. It is the editor's role to interpret the intended meaning where possible or decide whether that question should be eliminated from the tabulation.

Coding for Closed Ends

Before the final version of the questionnaire is printed, all closed-end structured questions are pre-coded. That means column numbers and code numbers are assigned to each answer area to facilitate transfer of data into a form that will be easy to enter into whatever computer program has been chosen to analyze the materials.

It is important that the transit marketing manager and the researcher discuss ahead of time how the data will be analyzed. Until the recent past, the usual practice was to keypunch computer cards which were then run through large mainframe computers. With the advent of large storage microcomputer hardware, and sophisticated data base and statistical analysis software packages, many researchers and/or marketers may be able to do the analysis using their own hardware/software, or that of their agencies. What's important is that the transit marketer and the researcher agree at the outset how the data will be handled.

Coding for Open-Ends

After all interviews are completed, coders determine into which of pre-established categories similar responses fall. A number is assigned to each category, the appropriate number written in the margin next to each response on every questionnaire, and the codes are available when data on the completed questionnaires are ready to be processed.

Transcription

Although tabulation can be done manually for simple questionnaires and small samples, most transit studies will use computerized statistical analysis.

Cleaning

While many checks can be made manually during the editing and coding stage, computers can more efficiently clean the questionnaire by checking the logical consistency of responses, the skip patterns or any other element of importance to the analyst. Computer cleaning requires the use of programs specially designed to clear, code, and reformat data. When errors occur in computer cleaning, they can be corrected mechanically by a programmed instruction, or manually by examining the original data.

5. Data Analysis and Interpretation

As mentioned earlier, data tabulation methods have changed radically in the last decade. Many marketers and researchers will now have the in-house capability of analyzing their own data. Again, though, what is important is that the transit marketer and his consultant or research supplier has available the appropriate tabulation programs and statistical packages. Since computer programs and microcomputer hardware vary in the amount of data they can accept and the statistical techniques they can execute, it is necessary that both the marketer and researcher have a thorough understanding of the size and complexity of the data base they need for their analysis.

The first step after the raw data have been prepared is to run marginals. Marginals are printouts which show how many people fall into the various answer categories and form a basis for deciding which sub-groups it will be possible, as well as enlightening, to analyze. The computer then counts the number of responses in each category for every question and prints these numbers on data tables. When selecting sub-groups for analysis, the objectives of the study should be kept in mind. For example, the objectives of a study may require the analyst to examine the attitudes and behavior of current riders, potential riders and non-riders.

After the analytical sub-groups have been determined and run, the data from each can be reviewed and compared for differences and/or similarities. If the questionnaire was effective, it will elicit information which will provide firm guidelines about whether or not to implement a proposed service or product change. As a rule of thumb, if 40% or more of the respondents indicate little or no interest in using the service or product being evaluated, it should probably not be implemented, unless changes desired by consumers can be made.

There are many statistical tests available for validiating the results of your sampling. One that is appropriate for some of the longitudinal and tracking research done in transit market research is the chi-square test. This is a statistical measure of the difference between the actual and expected results in your survey or a particular wave of your research. Put most simply, it is a way to check whether or not the degree of difference between what was expected (or what was found in earlier research) and what was found is significant. It is important that the transit marketer make certain that the researcher is performing the correct statistical analysis on the data and then validating the results.

Part 4: Professional Research Services

A. Guidelines for Selecting a Market Research Supplier

Because both the design and execution of research studies require a considerable degree of experience, skill and technical expertise, transit operators and other marketers without their own research departments should seek the services of marketing research professionals.

Professional aid is available from a variety of suppliers. Universities represent one resource. In addition, several different kinds of independent companies maintain qualified research personnel, such as:

- * Companies that specialize in market research, called research "suppliers."
- * Companies that specialize in marketing/communications with internal research capability, e.g. advertising agencies.

Marketers can find help in locating market research companies by using sources such as the American Marketing Association's *International Directory of Market Research Companies and Services*, an annual listing of most major research companies with addresses, telephone numbers and a brief synopsis of their services. (This is referred to in the trade as the "Green Book.") In addition, the American Marketing Association maintains offices in many American cities. You may be able to get lists of appropriate companies in your area by contacting the local A.M.A. chapter.

When selecting a market research firm or other organization for assistance, the transit marketer should make sure that the firm provides the specific services needed. While some firms provide the full range of services for the planning and execution of custom research studies, from consultation on the nature of the problem, through preparation of a proposal and actual execution and analysis of the study, others provide only very specialized services such as:

- * Qualitative research
- * Media research
- * Tracking or assessment research
- * WATS telephone interviewing
- * Door-to-door interviews
- * Computer output

The transit manager can use any or all of the services that may be offered by such a market research company, from questionnaire design to data analysis. The transit manager should present a written statement of the services that will be required, i.e. the number of completed interviews needed, estimated incidence of qualified respondents, sample selection, questionnaire design and production, field work, data processing, analysis and final report. The supplier should provide a detailed cost estimate listing each service to be provided and the approximate cost for each service. A contingency price variation of $\pm 10\%$ is normally included. As with all major purchases, it's probably best to shop around and get more than one estimate for what you need. It's also advisable to check with transit agencies in comparable markets to see what such services have cost them.

The following checklist can be followed as a guide in selecting a competent and reputable research company. It details the relevant areas to be considered in evaluating suppliers of market research.

MARKET RESEARCH SUPPLIER CHECKLIST

1. Study Design

- * Will supplier suggest alternative research designs if there is a better way to solve the problem?
- * Is supplier responsible for layout of questionnaire?
- * Is questionnaire easy to follow?
- * Does supplier write interviewer instructions? Are they clear?
- * Is physical set-up conducive to good work? Is all essential equipment available?
- * Is office well-organized so that components of a study can be found at any time?

2. Sample Selection

- * Does supplier advise on type of sample to use? Is a reliable probability sample obtained when desired? Are quotas met with quota sample desired?
- * Who draws sample, and how is sample drawing actually done?
- * Are sampling instructions clear and specific?
- * What is the procedure for callbacks?
- * What checking is done to make sure that interviewer followed sampling plan?

3. File Control and Supervision

- * What ensures a smooth work flow to field? Within field? And back? How can one detect problems?
- * What communications exist between home office and field? How does one know that all materials are mailed to field and received by field, and then returned when work is completed?
- * How would potential problems be resolved?
- * Do supervisors have clearly defined responsibilities? Can they adequately manage their interviewers? Do they validate and edit?
- * How does the company check or evaluate the supervisor's work? Interviewer's work?
- * What kind of training do supervisors and interviewers have?
- * Are briefing sessions and practice interviews conducted?

4. Validation

* What proportion of interviews are validated? Does supervisor validate? Does home office? What records are kept?

5. Editing and Coding

- * Who does the editing, and what type of editing do they do?
- * Are editing instructions clear?

- * For open-end questions, how are codes built? How large a sample of questionnaires are used, and how are they selected?
- * Who approves codes?
- * What controls ensure that each questionnaire is actually coded, coded properly, and coded only once? How are questionnaires identified? Individually? In batches? Is coding verified? How much?

6. Transcription

- * How accurately is data transferred to the computer?
- * Is typing or keypunching verified? How much?
- * Are data cleaned? Are cleaning instructions clear?

7. Tabulation

- * Who decides what tabulations shall be run?
- * Can you request additional tabulations after data is analyzed? At what cost?
- * Who writes tab specifications?
- * How are print-outs checked to ensure no mistakes were made?
- * How long will data be stored?

8. Analysis

- * Is analysis thorough? Are all objectives addressed? Are all appropriate subgroups examined? Are all necessary and appropriate special analyses conducted? Are appropriate statistical tests run when necessary?
- * Will the analyst go over data and findings with you?
- * Can the analyst explain data in lay people's terms?
- * Are written reports well-organized, concise, and clear?
- * Are recommendations reasonable? Can they be implemented?

B. Communication Between Marketer and Research Supplier

The key to a successful research project is full communication between marketer and supplier.

The transit marketing manager needs to prepare the following:

- * BACKGROUND MATERIAL & PROBLEM STATEMENT—the communication must start with a clear statement by the marketer of the problem and the kind of decision that will be based on the outcome of the research.
- * TIME AND COST REQUIREMENTS—these determine the size and scope of the study. Where these limitations are inconsistent with the type of study required for the marketing problem, the knowledgeable supplier will help to re-define the problem or suggest alternative approaches.
- * SAMPLE REQUIREMENTS, SAMPLE DATA ANALYSIS, AND PRESENTATION REQUIREMENTS—these help to determine the specific design elements of the study.

The supplier needs to prepare the following:

During the initial communication process, the supplier is responsible for preparing a proposal which outlines the supplier's understanding of the problem and specifies the entire research plan. The major components of the proposal are these:

- * STATEMENT OF OBJECTIVES—covering the marketing objectives and the research objectives.
- * STATEMENT OF DECISION CRITERIA, wherever possible.
- * SPECIFICATIONS OF THE SAMPLING PLAN, including:
 - a definition of the universe to be sampled (including geographic boundaries, if relevant);
 - a definition of the eligible respondents (i.e., the types of consumers who will be included and excluded from the study);
 - the number of people to be interviewed, and whether quotas will be established for consumer sub-groups of particular interest;
 - a description of methods used to draw the sample (full probability sampling, or census data, telephone directories, etc.)
- * OVERALL STUDY DESIGN (methodology).
- * OUTLINE OF INFORMATION TO BE OBTAINED, categorized by topic.
- * RELEVANT 'LOGISTICAL' CONSIDERATIONS, such as how products or ads will be placed with or exposed to respondents, whether each interviewer will handle all test items, how the test items will be identified, etc.
- * COST OF THE RESEARCH.
- * TIMING.

Where several suppliers have been approached for competitive bids on a study, each supplier's proposal, along with his actual bid, is the basis for deciding which will perform best. In the marketing research field, giving one supplier's proposal to competitors for purposes of bidding is highly unethical, unless the supplier is told in advance of preparing the proposal and reimbursed for the time spent on it.

C. Determinants of Research Costs

The following is a list of the major variables influencing the total cost of a research project:

- * The larger the sample, the higher the cost.
- * If only 10% of the population qualifies for the interview, the cost will be higher than if 80% qualify.
- * Personal, in-home interviews are always more costly than telephone interviews because of interviewer travel time.

- * Since interview length determines both the daily interview completion rate and the amount of data which must be analyzed, the longer the interview, the higher the cost.
- * The more open-end questions, the higher the coding costs.
- * Computer time and analytical staff time increases with the number of sub-groups analyzed.
- * Report writing/presentation requirements will increase cost.
- * If the study must be done in an unusually short time frame, overtime costs could be high.

When conducting a research study, however, one must consider value as well as cost. For example, although longer interviews cost more, they may be the only means of providing all required information. If it is not possible to acquire the needed information by telephone, it would be more economically beneficial to conduct in-home interviews. Telephone interviews which do not yield enough data upon which to base sound marketing and expenditure decisions are worthless.

One last note: it is becoming increasingly standard to run what are called longitudinal studies. These are essentially tracking studies over a longer period of time. In order to ensure that these are both efficient and effective, the marketing manager must have both a long and short range survey plan. As noted earlier, keeping surveys in as standard a form as possible, asking questions essentially similar to those asked in surveys of comparable public transit markets, making sure that questions asked over time remain the same and that the core of respondents remains demographically the same—all of these steps will help give your research greater reliability and projectability.

Bibliography of Marketing Research Textbooks

Backstrom, Charles. Survey Research. New York, New York: John Wiley & Sons, Inc., 1981.

- Benson, Dennis K., and Benson, Jonathon C. A Guide to Survey Research Terms. Columbus, Ohio: Benchmark Program, 1975.
- Boyd, Harper W., Stasch, Stanley, and Westfall, Ralph. *Marketing Research, Text and Cases.* Homewood, Ill.: Richard D. Irwin, Inc., 1977.
- Bruno, A. J. "Validity in Research: An Elaboration." Journal of Advertising Research. June 1975.
- Bradburn, Norman and Sudman, Seymour. Asking Questions. San Francisco, California: Jossey-Bass Publishers, 1982.
- Bywaters, David R. and Hamilton, Richard A. *How to Conduct Association Surveys.* Washington D.C.: Foundation of the American Society of Association Executives, 1976.
- Cox, Keith K., & Enis, Ben M. *The Marketing Research Process.* Pacific Palisades, Calif.: Goodyear Publishing, 1972.
- Edwards, Allen L. *Experimental Design in Psychological Research.* New York: Holt, Rinehard and Winston, 1972.
- Ferber, Robert. Handbook of Marketing Research. A publication of the American Marketing Association, 1975.
- Festinger-Katz. Research Methods in the Behavioral Sciences. New York: Holt, Rinehard and Winston, 1954.
- Fleishman, Daniel. Route Specific Transit Marketing In Minneapolis/St. Paul. Washington D.C.: UMTA, 1985.
- Gigante, Lisa & Koo, Emily. Marketing Public Transit. Lansing, Michigan: Michigan DOT, 1985.
- Glazer, G. J. and Metzger, G. D., "National Estimates of Non-listed Telephone Households and Their Characteristics." *Journal of Marketing Research.* August 1975.
- Glazer, G. J. and Metzger, G. D., "Random Dialing as a Method of Telephone Sampling." *Journal* of Marketing Research. February 1972.
- Haynes, J. B. and Roth, J. T. "Competitive Bidding for Marketing Research Services: Fact or Fiction?" *Journal of Marketing.* July 1974.

Holbert, P. "How Managers See Research." Journal of Advertising Research. December 1974.

Hughes, G. D. Attitude Measurement and Marketing Strategies. Glenview, Illinois: Scott Foresman, 1971.

- Jordan D., Arnstein, S., Metcalf, E., *Effective Citizen Participation in Transportation Planning.* Vol. II, Washington D.C.: FHWA, 1976.
- Joselyn, Robert W. Designing The Marketing Research Project. New York: Petrocelli/Charter, 1977.
- Kess, George. Marketing Research. Reston, Virginia: Reston Publishing Company, Inc., 1979.
- Kotler, Philip. Marketing Management: Analysis, Planning and Control. Englewood Cliffs, N.J.: Prentice-Hall Inc., 1972.
- Mayer, Charles S. "How to Choose a Research Supplier" *Journal of Marketing Research.* Vol. V May 1967, pp. 135-138.
- McNemar, Quinn. Psychological Statistics. New York: John Wiley and Sons, Inc., 1969.
- Parket, I. Robert. Statistics for Business Decision-Making. New York: Random House, 1974.
- Parten, Mildred. *Surveys, Polls and Samples.* New York: Cooper Square (affiliated with 50 Fourth Avenue), 1950.
- Retzlaff, Jeffrey. Soucie, Kevin and Beimborn, Edward. Use of Market Research in Public Transit. Washington D.C.: UMTA, 1985.
- Ross, Peter, Anderson, Andy B., and Wright, James D. *Handbook of Survey Research.* New York, New York: Academic Press, 1983.
- Ruddick, Morris, Sherwood, Philip, and Stevens, Robert. *The Marketing Research Handbook.* New York, New York: Prentice Hall, Inc., 1983.
- Twedt, D. "Authorization, Control, and Evaluation of Marketing Research Projects." *Journal of Marketing Research.* February 1975.
- Tull, Donald S. & Albaum, Gerald S. *Survey Research: A Decisional Approach.* New York: Intext Educational Publishers, 1973.
- Warwick, Donald P. & Lininger, Charles A. *The Sample Survey: Theory and Practice.* New York: McGraw-Hill, 1975.
- Wentz, Walter B. Marketing Research: Management and Methods. New York: Harper & Row, 1972.

TECHNICAL APPENDIX A: Guidelines for Determining Sample Size

Sampling is based on laws of probability which state that the larger the sample size, the more confidence can be placed in the findings. This confidence is measured statistically in terms of a range of error that is likely to occur due to random chance alone.

In designing studies, it is important to estimate the error range in advance in order to determine the sample sizes that are required. Since costs rise as sample size increases, it is necessary to determine a reasonable trade-off between the desired accuracy of the results relative to the funds available for the study. It is also important to consider the number and size of the analytical sub-groups in the sample in order to be sure that results can, in fact, measure meaningful differences. In this respect, it may be necessary to rank the sub-groups by their value to the final analysis and to eliminate certain ones from the study altogether.

Sample size will depend on several factors: the purpose of the research, the nature of the sample design, and the degree of diversity within the target population. Sample size will also vary according to the degree of desired precision and confidence in the results of the study. Available time, money, and personnel are also key considerations in determining sample size. Below are a few general rules regarding sample size:

- 1. The larger the sample size, the less probability of error.
- 2. The more sub-groups you need analyzed, the larger your sample needs to be.
- 3. Conversely, the greater the similarities within the study population, the smaller the sample can be.

The sample needs to be large enough to ensure an acceptable level of precision. The researcher can never be 100% sure that the data are dependable. However, he or she needs to get the maximum level of confidence within the bounds established by available resources, so that the information resulting provides a good basis for decision-making. Researcher and marketer thus become involved in a trade-off between various cost and time constraints weighted against the sample size and the level of detail in the analysis. If the amounts of time and money are too severely limited, the whole idea of gathering primary data should be reconsidered.

There are several formulas for computing sample size available to experienced researchers. One table is presented below as an example:

	Confidence Levels											
Tolerated Error	95 Samples in 100	99 Samples in 100										
1%	9,604	16,587										
2%	2,401	4,147										
3%	1,067	1,843										
4%	600	1,037										
5%	384	663										
6%	267	461										
7%	196	339										

SIMPLE RANDOM SAMPLE SIZE FOR SEVERAL DEGREES OF PRECISION

From the above it is evident that a great price is paid for increased precision and confidence. Although increasing the sample size lowers tolerated error and/or raises confidence limits, the improvement is not proportional to the increase in sample size required.

Two Samples or Groups

The issue in the two-group case is to determine whether the difference between the results is statistically significant, i.e., are the differences on the mean percentages between two groups greater than would be expected due to random chance.

The following table is used to help one determine whether one is designing the study with samples large enough to measure differences between two groups at the 80% confidence level. The table provides the minimum range or difference between the group means that will be required before the results can be considered greater than those occurring due to chance differences.

	Av	verage of the	e Two Obse	rved Percer	nts
Group Sample Size	10% 90%	20% 80%	30% 70%	40% 60%	50% 50%
100	5.4	7.2	8.3	8.8	9.0
200	3.8	5.1	5.8	6.3	6.4
300	3.1	4.2	4.8	5.1	5.2
400	2.7	3.6	4.1	4.4	4.5
500	2.4	3.3	3.7	4.0	4.0
750	1.9	2.6	3.0	3.3	3.2
1000	1.7	2.2	2.6	2.8	2.9
2000	1.2	1.6	1.9	1.9	2.0

MINIMUM RANGE REQUIRED TO BE 80% CONFIDENT THAT THE PERCENT FROM TWO GROUPS ARE DIFFERENT

. .

. .

For example, if one were to do a tracking study of 100 per wave and expected to get about a 5 percentage point difference in ridership, one would want to know whether this increase was large enough to be sure that it was the result of our marketing program and not a chance occurrence due to the sample size one is using. Using the assumption of 20% ridership level, a sample size of 100 would require a difference of 7.2 percentage points before results would be statistically significant at the 80 percent confidence level. Since one wants to be able to measure a 5 percentage point difference in ridership between two waves, we would need to increase the sample size to 200 per wave. At this size, we would be able to measure a 5.1 percentage point difference.

An illustration of two groups from the same sample would be a case where one wants to compare the attitudes between the riders on two routes toward their service to determine if the percent liking the service differs. Suppose it is decided that a 6 percentage point difference between liking the service is acceptable, one needs to know what sample size is needed to indicate that a 6 percentage point difference represents a real difference in consumer attitudes toward service. Assuming that about 70 percent on the average will like the service, the table indicates that a sample size of 200 on each route is necessary to measure a 5.8 percentage point difference between routes.

Setting Levels of Confidence

In the above discussion, an 80 percent confidence level, or about four to one odds of making the right choice, has been used. However, for situations of a *very high risk* nature, a confidence level of 90% or 95% may be worth the additional cost of increasing sample size. In fact, when any important decisions are to be made, it is probably necessary to achieve a 95% level of confidence, with a tolerated error ranging from + or - 3% to + or - 8%. Conversely, in certain situations where risks are minimal, a lower confidence level, e.g., 66 percent or 2 to 1 odds, may be all that is required.

In the final analysis, the confidence level to select is based on the risk involved versus cost considerations. Confidence limits for any level can be obtained from formulas in statistical textbooks.

TECHNICAL APPENDIX B: Strengths and Weaknesses of Interviewing Methods

The following pages discuss the pros and cons of the various interviewing methods.

In-Home, Personal Interviews

The major advantages are:

- permits selection of highly representative and projectable samples
- permits interview of greater length
- makes it possible to incorporate small visual aids (cue cards, scaling cards, etc.) into the interview
- provides maximum opportunity for probing open-end responses

The major disadvantages are:

- high cost
- vulnerability to geographical "cluster effects" unless a sufficient number of areas are included
- requires highly experienced research personnel
- difficulty of getting interviewers to go into "undesirable" neighborhoods

Central Location/Street Intercept Interviews

The major advantages of the personal interview conducted in a central location are:

- possible to incorporate visual aids into the interview, such as videotaped commercials which cannot be carried door-to-door
- permits interview of greater length
- saves travel time and permits group interview

The major disadvantages are:

 less sampling control—impossible to draw a probability sample, possibility of under-or-over representing certain consumer groups

On-Board Surveys

The major advantages are:

- very inexpensive method for interviewing transit users
- provides a good cross-section of riders on specific routes

In addition to the disadvantages cited for central location/street intercept interviewing, further disadvantages of on-board surveys are:

- cannot survey non-riders
- limited amount of information obtainable
- possible bias due to on-board environment
- limited questionnaire design if self-administered

Telephone Interviews

The major advantages of telephone interviewing are:

- relatively low cost per interview
- easier sample selection procedures
- no geographical cluster effects
- easier to reach certain population sub-groups (such as those in very high income brackets or those in less desirable neighborhoods)
- less expensive call-backs
- efficient for large-scale screening to locate low incidence consumer groups

The major disadvantages are:

- lack of opportunity to present visual stimuli
- less flexible than personal interviews in terms of interviewing techniques that require respondent involvement
- limitations in interview length
- non-coverage or difficulty in covering certain population groups (non-telephone households)

TECHNICAL APPENDIX C: Focus Groups

Focus group discussions are becoming increasingly popular in qualitative transit market research. This technique consists of a structured discussion with a group of carefully selected people from the market segment being studied. Talking with people in groups rather than individually is an approach designed to encourage spontaneous interchange. The technique is based on the notion that people, allowed to be spontaneous in a social, relaxed, non-threatening environment, will reveal personal opinions that are not captured in direct questioning. In fact, some experienced facilitators contend that the dynamics of an in-depth group interview prompts respondents to reveal things of which they themselves were previously unaware.

Scope

The groups can involve anywhere from 10 to 14 people led by a facilitator. It is crucial that the objectives be thoroughly understood before the discussion is conducted. Whether the discussion is being used to elicit information, probe attitudes, resolve issues, develop a survey, or build support or consensus, the objective sets the direction for the facilitator who seeks the information in a style appropriate to the occasion. Attitudinal information may require the preparation of a list of questions by the facilitator to serve as a guide for discussion. As a result, the information gathered is generally greater than the sum of the individual respondents' comments. The interaction itself can provide findings otherwise unattainable which, when summarized, yield insights into motivations and priorities, the salience of various issues and possible negotiating points to be considered in the implementation of alternative programs. These sessions are often tape recorded to provide more freedom to the facilitator to lead group members, without having to be concerned with "getting everything down."

Selection

Successful focus groups require that everyone involved attend the discussion and be on time. Often those conducting the research offer some kind of financial incentive to ensure punctual attendance. It is advisable that participants be telephoned the day prior to the discussion to be reminded of the time and place. The composition of the group is very important. The selection process must take into account the relevant market segments to be studied. It is very important to take caution not to over or under represent any significant segment of the population to be sampled. Researchers must work to reduce the chance of conducting all the group discussions with respondents holding similar viewpoints, and thus biasing the projected reaction of the whole community.

Training

It is very important that the group facilitator or moderator work closely with the marketing manager ahead-of-time so that he or she understands exactly what kind of information is being sought. Despite its "casual" setting, focus group discussions are pieces of serious research and should be treated as such. Three to six weeks should be allotted to the preparation for group discussions. During this period a list of relevant questions is developed and refined and the types of panel respondents are chosen. Other steps in the process include: developing and mailing any written notices to be used, developing the discussion guide, scheduling the time for the group to meet and reserving the facilities. The actual discussions generally last about 2 to 21/2 hours, unless the respondents are highly enthusiastic and prefer to keep going. The time required for

analysis of the group interviews depends on the number of groups conducted, degree of detail required, and whether or not each group is to be summarized separately.

Summary

The group process itself is interactive. The results depend upon the selection of the facilitator, the selection of the respondents, the objective of the discussion, the preparation and the flexibility and adaptability of the research agenda. Since this approach is qualitative and is not statistically projectable, it is most effectively used in conjunction with other quantifiable techniques.

TECHNICAL APPENDIX D: Survey of Surveys

In May of 1983, the Center for Urban Transportation Studies at the University of Wisconsin– Milwaukee conducted a survey of transit agencies to examine their use of surveys as a market research tool. This survey consisted of a questionnaire mailed out to a preselected sampling of transit agencies. The transit agencies to be chosen as potential respondents were identified in the American Public Transit Association publication PASSENGER TRANSPORT as having conducted a market research study during 1982 and 1983. Forty of the 60 transit agencies selected for the survey responded.

The questionnaire solicited information from each transit agency concerning the purpose of the survey, the method by which the survey was administered, usefulness of the information collected by the survey, and what, if anything, would have been or will be changed if the survey was/is conducted again. Each respondent was asked to include a sample of the survey form when returning the questionnaire. Forty-seven sample surveys were received from the 40 transit agencies responding; some agencies including more than one survey, and others completed only the questionnaire.

1. By reviewing the responses to the questionnaire, some general observations on the use of the surveys by the various transit agencies can be made.

User/Non-User and Trip Profile

- to obtain information on the characteristics of user and non-user (i.e., demographic, socioeconomic, etc.);
- to obtain origin-destination and other trip characteristic information (i.e., frequency of ridership, method of payment, etc.).

User/Non-User Attitudes

- to obtain information necessary to assess user and non-user attitudes toward and perceptions of the transit agency, transit service, etc.;
- to identify inhibitors and potential incentives to using the transit service.

Marketing Effectiveness

- to measure ridership awareness of advertising campaigns and evaluate marketing campaigns;
- to obtain information necessary to develop a strategy to increase ridership;
- to determine the effect of various marketing techniques on the usage of the transit system.

Economic Impacts

- to determine the economic impacts of the transit system on the surrounding community.
- 2. Almost all of the sample surveys received had more than one stated purpose or objective. The practice of "piggybacking" or combining questions onto one survey to serve more than one purpose is a common element found in all the sample surveys received.
- 3. The most common method of conducting a survey, as seen observed in the "survey of surveys" is the use of an on-board survey. Approximately 49 percent (23 surveys) of the surveys received were on-board surveys. Telephone interviews made up 38 percent (18 surveys) of those received. Personal interviews (on-board interviews, mail-intercept, inhome) comprised 10 percent (5 surveys) of the total received, with mail-out surveys comprising 2 percent (1 survey) of the total received.
- 4. Seventy-five to 80 percent of the sample surveys received having "user/non-user profiles" and "user/non-user attitudes" as their primary purpose were conducted with an on-board survey. The remaining 20 to 25 percent were conducted with a telephone interview.

Eighty percent of the sample surveys received having "marketing effectiveness" as their primary purpose were conducted with a personal interview. The remaining 20 percent were conducted with an on-board survey.

The one economic impact survey received was conducted through an on-board, personal interview.

- 5. Some of the general comments made concerning changes that might have been or will be made to the survey itself include:
 - include questions seeking information necessary to conduct a "follow-up" (i.e., name, phone number)
 - include more questions concerning the attitudes, perceptions, and opinions of both users and non-users;
 - develop or improve coding of questions to aid in tabulation.

(from Use of Market Research in Public Transit)

EXHIBIT A Sample Screener Questionnaire

			Time Interview Started:									
Start 79- 80-												
(Please F	Print)											
Name: _			_ Tel.#()									
Address:			_ County:									
City/State	e:			Zip:								
Interview	er:			Date:								
Name of	Nearest Main St			Sampling Area:								
	y name is ease speak to the man or w											
•	If neither the male nor fen another adult, 21 years of a			s at home, ask to speak to								
•	lf no adult is at home, find o call back then.		of the head(s)	of household will return, and								
	Time for call back:	AM PM										
			RESPONDE	NT IS: Male Head								

We are conducting an opinion poll about various public service issues in your community, and your household has been randomly selected as part of our sample. The opinions of your household will be very valuable in defining the needs and attitudes of your neighborhood, and will be kept in the strictest confidence.

- A. First, what do people call the section of town where you live? _
- 1(a) People of different ages often have different points of view about public service issues. Therefore, we need to known the name and age of each person presently living in the household. Let's start with the male head of household. What is his name and age? (Record in appropriate spaces on referral sheet. If no male head of household, write in "none.")
- 1(b) Now, please tell me the names and ages of any other males, 16 years of age or older, who presently live in your household. Start with the oldest person and end with the youngest. Do not mention students who life outside the ______ Metropolitan Area during the school year. (Record in appropriate spaces on referral sheet. Remember to obtain both name and age. If respondent is male, but not household head, make sure he includes himself.)
- 1(c) And what is the name and age of the female head of household? (Record in appropriate spaces on referral sheet. If no female head of household, write in "none.")
- 1(d) And what are the names and ages of any other females, 16 years of age or older, who presently life in your household? Again, start with the oldest person and end with the youngest. Do not mention students who live outside the ______ Metropolitan Area during the school year. (Record in appropriate spaces on referral sheet. Remember to obtain both name and age. If respondent is female but not household head, make sure she includes herself.)
- 1(e) Of **all** the household members you've mentioned, which of them have jobs outside your home? [Circle as many as apply in column 1(e) on referral sheet.]
- 1(f) Which, if any, of the household members living at home, are students? (For each person named ask:) Does (household member) go to a high school, college or technical school? [Circle in column 1(f) on referral sheet.]
- 1(g) Which of your household members possess a driver's license? [Circle as many as apply in column 1(b).]
- 1(h) Which of the people in your household make a round trip beyond walking distance of home **at least twice a week**? [Circle as many as apply in column 1(h). If none, **terminate**.]
- 1(i) [For each household member circled in column 1(h), ask:] Is (household member) at home right now? [Record in column 1(i).] Now I have to ask just one more question about your household that will permit us to group your responses with those of other people with similar characteristics.

2. Is the total income in your hosuehold *under* \$10,000, or is it \$10,000 or more, before taxes? Please include the income of *all* your household members.

Under \$10,000—	Is that: Under \$7,500 1 or \$7,500 or more 2	Under \$5,000 3 or \$5,000 or more 4
 \$10,000 or more—	- Is that: Under \$15,000 5 or \$15,000 or more 6	Under \$20,000 7 or \$20,000 or more 8

Refer to Q.1 on referral sheet to determine the household member with whom you should continue. Select respondent from those at home now [column 1(i)] according to priority list below. For example, if a college student and a working female are at home, you'd interview the college student. If a working male and a working female are at home, you'd interview the working male, and so forth.

Select in this priority:

- 1. A college student
- 2. A high school student
- 3. A working male
- 4. A working female
- 5. A non-working male, not a student
- 6. A non-working female, not a student

If no other household member at home, continue with same respondent. Circle respondent to be interviewed on Q.1 referral sheet (right-hand column).

If you are to continue with same respondent, skip to Q.3(b).

If you are to interview another household member ask: May I please speak with (household member)?

Continue with full questionnaire with proper respondent.



EXHIBIT B Sample Full Questionnaire

	Time Interview Started:									
(Please Print)										
Name:										
Address:	County:									
City/State:		Zip:								
Interviewer:		Date:								
Name of Nearest Main St		Sampling Area:								

1. *Introduce yourself with:* Hello, I'm ______ from the *(insert company name).* We're conducting a survey to find out how people in the Metropolitan Area feel about various public service issues and we need your opinion about the needs of your community.

Continue with all respondents:

2. There are many problems and social issues affecting the quality of life in our urban areas today. I'll read you a list of such issues, and I'd like your opinion about the *importance* of each one to the Metropolitan Area.

To help you express your opinion, I'd like you to write down a few short phrases and read them back to me. Do you have a pencil and paper? (*Wait until respondent has writing materials.*) Now, please list the following phrases, one under the other: Extremely, Very, Quite, Rather, Not Too, and Not At All.

Record whether respondent wrote down phrases:	Yes	1
	No	2

First, think about the issue of *(read item X'd)*. Using the phrases you just wrote down, would you say this issue is **extremely** important, **very** important, **quite** important, **not too** important, or **not at all** important to the overall welfare of the Metropolitan area? *(Record below.)*

And how important is *(read next item)* to the overall welfare of the Metropolitan area? (Record below and repeat question for each item on the list. Read down, then return to first item until you have reached your starting point.)

If respondent did not write down rating phrases, repeat the phrases for each item by saying: "Would you say this issue is **extremely** important, etc."

Star	t	IMPORTANCE RATINGS												
Here	ltems	Extremely	Very	Quite	Rather	Not Too	Not At All							
() Redeveloping the downtown business district	6	5	4	3	2	1							
() Relief of traffic congestion	. 6	5	4	3	2	1							
() Controlling air pollution	. 6	5	4	3	2	1							
() Providing a good bus system for public transportation	. 6	5	4	3	2	1							
() Providing the needs of senior citizens	6	5	4	3	2	1							
() Installation of the rapid transit system	6	5	4	3	2	1							

Because it would take up too much time to cover all these issues in one interview, we are asking each person to talk about only one of them. In your household, we'd like to cover issues relating to public transportation, especially the metro area bus system.

3. What is the **name** of the bus system in your area? (Write in verbatim what respondent says on line below.)

(write in name of bus system)

4. How would you rate the overall quality of service provided by the bus system in the ______ Area? Would you say it is excellent, very good, good, fair, not so good, or poor? (*If respondent says ''don't know/never ride,'' ask:*) Based on whatever you know about the bus system or what you've seen of the buses, what is your impression of the bus service? Is it . . . (*Read list*):

Excellent															6
Very Good															
Good						•			•		•	•	•	ÿ.	4
Fair	•	•	•	•	•	•	•	•			•	•	•		3
Not So Good			•	•				•				•			2
Poor													•		1

5(a)	Is there a bus line within walking distance	of your home?	
		Yes [ASK Q.5(b)-(c)]	1
		No [SKIP to Q.6]	2
		Don't know	3

5(b)	Do you know where	the bus goes, either for all or part of its route	?

Yes	0		÷	•	•	•	•		•	•				•	•			1
No	1	•				•										,		2

5(c) Do you know the location of the bus stop closest to your home?

Yes	÷													1
No .				•	•								•	2

6. On average, how often do you, yourself, use the Metro Area bus system in a typical month? Count the bus ride from home to your destination as one trip, and the bus ride from your destination to home as a separate trip. Now, about how many bus trips do you make in a typical month?

(write in number of trips)

[If less than 3 trips in Q.6, ask Q.7. if 3 trips or more in Q.6, skip to Q.8(b)]

7. Let's suppose that new and improved bus service were introduced in your area to match your travel needs. Use the phrases you wrote down before to tell me how likely you would be to use the bus then. Would you be extremely likely, very likely, quite likely, rather likely, not too likely, or not at all likely to use the bus?

	[GO TO Q.8(a)] ———	Extremely 6 Very 5 Quite 4 Rather 3
	[SKIP TO Q.8(b)] ———	Not too
8(a)	Do you usually walk to work, or do you u	se some other form of transportation? Walk [SKIP to Q.9(a)]
8(b)	Do you work part-time or full-time?	Part-time
8(c)	Do you work a regular Monday to Friday	work week?
	Yes 1	No 2 Schedule 3
	Do you travel to and from work	No 2 Schedule 3 How many days a week
	Do you travel to and from work (Read List)	No 2 Schedule 3
	Do you travel to and from work	No 2 Schedule 3 How many days a week do you work? <i>(Write in Number)</i> How many of those days are
	Do you travel to and from work (Read List) During morning and evening	No 2 Schedule 3 How many days a week do you work? (Write in Number)
	Do you travel to and from work (<i>Read List</i>) During morning and evening rush hours 1	No 2 Schedule 3 How many days a week do you work?

(If respondent works on weekdays ask:) On the weekdays when you work do you travel to or from work: (Read List) Out of your last 10 work trips, how many times did you use the bus? (Write in Number)

During morning or evening rush hours 1 During daytime non-rush hours 2

 Nights
 3

 Or does your schedule vary
 4

8(d) [If work trips were less than 6 times in Q.8(c), ASK Q.8(d), otherwise SKIP to Q.9.] If new improved bus services were introduced in your area to match your **working** trip needs, how likely would you be to use the bus more often for work trips? Would you be extremely likely, very likely, quite likely, rather likely, not too likely, or not at all likely to use the bus more often?

(Record answer on referral sheet.)	LIKELIHOOD: 6 Extremely 5 Very 5 Quite 4 Rather 3 Not too 2
	Not at all

- 9. Refer to referral sheet:
 - If respondent does not work, SKIP to Q.11.
 - If respondent works and has used the bus 6 or more times [in Q.8(c)] **or** would be extremely-rather likely to use the bus more often for work [Q.8(d)], ASK Q.9 and Q.10.
 - If respondent works and used the bus less than 6 times in Q.8(c) and would **not be** likely to use the bus more often for work trips [Q.8(d)], SKIP to Q.11.

Now, I'd like to have your overall opinion of car and bus transportation as a means of getting to and from work. I'd like your opinions of **both** types of transportation, regardless of how much or how little you now use either type. If you don't drive a car to work, or if there is no bus service from neighborhood to where you work, just base your opinion on what you **think** travel in the Metropolitan Area would be like if you had a car or a bus line at your disposal.

Use the list of phrases you wrote down before, with "extremely" at the top and "not at all" at the bottom—to indicate how satisfied you are overall with the car and the bus.

- 9(a) First, let's talk about car transportation. On an overall basis, how satisfactory would you say *driving a car* is as a means of getting to and from work? [If respondent did not write down rating scale, read list below. Record response in column 9(a).]
- 9(b) And on an overall basis, how satisfactory would you say *riding a bus* is as a means of getting to and from work? [Record below in column 9(b).]

	Q.9(a)	• •
Overall Satisfaction	Driving a Car	Taking a Bus
Extremely	. 6	6
Very		5
Quite		4
Rather	. 3	3
Not too		2
Not at all	. 1	1

10. Now, I'd like to know how satisfied you are with car and bus transportation—as they exist at present—in terms of some specific considerations. I'll read the list of statements and as I read each one, tell me first what you think of driving a car to work, and then what you think of riding a bus to work, in terms of that consideration. Give me your opinion of both driving a car and riding a bus, regardless of the transportation you now use for your work trip.

Use the list of phrases with "Extremely" at the top and "Not At All" at the bottom to express your opinion.

Now, how satisfactory is "*driving a car—to work*" in terms of: (*Read X'd item. Record response in appropriate row. If respondent did not write down rating phrases, repeat phrases by saying:*) Would you say it was extremely satisfactory, very satisfactory, quite satisfactory, etc.

And how satisfactory is "riding a bus—to work" in terms of: (Read X'd item. Record response in appropriate row. Repeat rating phrases if necessary.)

(Continue in same manner for all items on the list. Ask for rating of car and then rating of bus on the same item before moving on to the next item.)

Start		How Satisfactory Is (driving a car)	SATISFACTION RATINGS											
Hei	re	(riding a bus) to Work in Terms of:	Extremely	Very	Quite	Rather	Not Too	Not At All						
()	Letting you come and go whenever you want to												
		Car Bus	6 6	5 5	4 4	3 3	2 2	1 1						
()	Eliminating worry about delays caused by traffic congestion												
		Car Bus	6 6	5 5	4 4	3 3	2 2	1 1						

Sta	art	How Satisfactory Is (driving a car)	SATISFACTION RATINGS										
He		(riding a bus) to Work in Terms of:	Extremely	Very	Quite	Rather	Not Too	Not At All					
()	Being an accepted way to travel by people like yourself											
		Car	6	5	4	3	2	1					
		Bus	6	5	4	3	2	1					
()	Letting you do your part on											
· ·	,	conserving gasoline											
		Car	6	5	4	3	2	1					
		Bus	6	5	4	3	2	1					
()	Being easy to use		Ū		Ũ	-						
(/	Car	6	5	4	3	2	1					
		Bus	6	5	4	3	2	1					
()	Being dependable in bad weather		U		U	-						
(,	Car		5	4	3	2	1					
		Bus	6	5	4	3	2	1					
()	Letting you do your part in	· ·	Ū		U	-						
`	,	reducing air pollution											
		Car	6	5	4	3	2	1					
		Bus		5	4	3	2	1					
()	Getting you from door to door	-	-	-	•	-						
(,	quickly											
		Car	6	5	4	3	2	1					
		Bus	6	5	4	3	2	1					
()	Letting you avoid having to fight	•			Ū.	-	•					
``	,	traffic											
		Car	6	5	4	3	2	1					
		Bus		5	4	3	2	1					
()	Being a comfortable way to travel											
,	.,	Čar		5	4	3	2	1					
		Bus	6	5	4	3	2	1					
()	Being an inexpensive way to travel											
,		Car		5	4	3	2	1					
		Bus	6	5	4	3	2	1					
()	Being reliable, with little											
		risk of delay											
		Car	6	5	4	3	2	1					
		Bus	6	5	4	3	2	1					
()	Letting you come and go											
		whenever you want to											
		Car	6	5	4	3	2	1					
		Bus	6	5	4	3	2	1					
()	Not having to worry about being											
		mugged while walking to or											
		waiting for the (car/bus)											
		Car	6	5	4	3	2	1					
		Bus	6	5	4	3	2	1					

11. Refer to Referral Sheet.

Now I'd like to have your overall opinion of car and bus transportation as a means of getting to and from the area where you prefer to shop. Remember that we are talking about the shopping you do other than grocery shopping. I'd like your opinions of **both** types of transportation, regardless of how much or how little you now use either type. If you don't drive a car for shopping, or if there is no bus service from your neighborhood to where you shop, just base your opinion on what you **think** travel in the Metropolitan Area would be like if you had a car or a bus line at your disposal.

Use the list of phrases you wrote down before, with "Extremely" at the top and "Not At All" at the bottom—to indicate how satisfied you are overall with the car and the bus.

- 11(a) First, let's talk about car transportation. On an overall basis, how satisfactory would you say *driving a car* is as a means of getting to and from the area where you prefer to shop? [If respondent did not write down rating scale, read list below. Record response in Column 11(a).]
- 11(b) And on an overall basis, how satisfactory would you say *riding a bus* is as a means of getting to and from the area where you prefer to shop? [Record below in column 12(b).]

	Q.11(a)	Q.11(b)
Overall Satisfaction	Driving A Ca	ar Taking A Bus
Extremely	6	6
Very		5
Quite		4
Rather		3
Not Too		2
Not At All		1

12. Now, I'd like to know how satisfied you are with car and bus transportation—as they exist at present—in terms of some specific considerations. I'll read the list of statements and as I read each one, tell me first what you think of driving a car for shopping, and then what you think of riding a bus for shopping, in terms of that consideration. Give me your opinion of both driving a car and riding a bus, regardless of the transportation you now use for your shopping trips.

Use the list of phrases with "Extremely" at the top and "Not At All" at the bottom to express your opinion.

Now, how satisfactory is "*driving a car for shopping*" in terms of: (*Read X'd item. Record response in appropriate row. If respondent did not write down rating phrases, repeat phrases by saying:*) Would you say it was extremely satisfactory, very satisfactory, quite satisfactory, etc.

And how satisfactory is "riding a bus for shopping" in terms of: (*Read X'd item. Record response in appropriate row. Repeat rating phrases if necessary.*)

(Continue in same manner for all items on the list. Ask for rating of car and then rating of bus on the same item before moving on to the next item.)

Start		How Satisfactory Is (driving a car)	SATISFACTION RATINGS										
He		(riding a bus) to Work in Terms of:	Extremely	Very	Quite	Rather	Not Too	Not At All					
()	Letting you come and go											
		whenever you want to	c	F	4	0	0	4					
			6	5	4 4	3	2 2	1					
,		Bus	6	5	4	3	2	1					
()	Eliminating worry about delays caused by traffic congestion											
		Car	6	5	4	3	2	1					
		Bus	6	5	4	3	2	1					
()	Being an accepted way to travel											
		by people like yourself											
		Car	6	5	4	3	2	1					
		Bus	6	5	4	3	2	1					
()	Letting you do your part on											
2		conserving gasoline											
		Car	6	5	4	3	2	1					
		Bus	6	5	4	3	2	1					
()	Being easy to use											
,		Car	6	5	4	3	2	1					
		Bus	6	5	4	3	2	1					
()	Being dependable in bad weather											
x	,	Car	-	5	4	3	2	1					
		Bus		5	4	3	2	1					
()	Letting you do your part in											
X	/	reducing air pollution											
		Car	6	5	4	3	2	1					
		Bus	-	5	4	3	2	1					
()	Letting you avoid having to fight											
X.	/	traffic											
		Car	6	5	4	3	2	1					
		Bus	-	5	4	3	2	1					
()	Being a comfortable way to trave	1										
(,	Car		5	4	3	2	1					
		Bus		5	4	3	2	1					
()	Being an inexpensive way to trave		-	1.5	-	_						
ι,	,	Car	-	5	4	3	2	1					
		Bus		5	4	3	2	1					
			-	-	-	-		-					

Sta	art	How Satisfactory Is (driving a car)	ON RATIN	TINGS					
He	re	(riding a bus) to Work in Terms of:	Extremely	Very	Very Quite Rat		Not Too	Not At All	
()	Being reliable, with little risk of delay							
		Car	6	5	4	3	2	1	
()	Bus Not having to worry about being mugged while walking to or waiting for the (car/bus)	6	5	4	3	2	1	
		Car	6	5	4	3	2	1	
		Bus	6	5	4	3	2	1	
	13.	Have you used (ITEM) in the past X'd item first and continue.)	month? <i>(C</i>	ircle be	elow unde		Q.13 Q.13 Used in ast Montl		
		Improved timetables for individual	bus lines .				1		
		A new brochure describing how yo	u use the b	ous sys	tem		2		
		New maps showing the bus routes	within the	Metrop	olitan Are	ea	3		
		New outdoor map displays of bus r	outes				4		
		Other (SPECIFY):					5		
		None					xxx		
	13(a	 To the best of your knowledge, do services or any other special bus 					cial expre	SS	

Yes [ASK Q.13(b)-(d)]	1
No	
Don't know/Not sure	
SKIP TO Q.14(a)	3

- 13(b) What specific services can you think of? (*PROBE:*) Any others? [*Circle below under column 13(b).*]
- 13(c) [For each item **not** circled in column 13(b), ASK:] Have you ever heard of (ITEM)? [Circle below under column 14(c).]
- 13(d) [For each item circled in columns 13(b) and 13(c), ASK:] Have you used (ITEM) in the past month? [Circle below under column 14(d).]

	Q.13(b) Heard of Unaided	Q.13(c) Heard of Aided	Q.13(d) Used in Past Month
Park 'n Ride Services	1	1	1
The Metro Flyer	2	2	2
Express service on regular bus routes	3	3	3
The Downtown Shuttle	4	4	4
Other (SPECIFY:)		XXX	5
None	0	XXX	XXX

14(a) Does the MTA have a special number you can call to get bus service.

Yes [ASK Q.14(b)]		•	÷			÷		•	1
No [SKIP to Q.15]									
Don't know/Not sure									

14(b) Have you ever used that telephone information services?

Yes [ASK Q.14(c)]	,	•	•	•	•	•	•		•	•	1
No (SKIP to Q.15) .									•		2

14(c) How would you rate the telephone information service, based on your experience using it? Would you say it was Excellent, Very Good, Good, Fair, Not So Good, or Poor?

Excellent									÷									1
Very Good	•				•	•									•			2
Good	•	ł																3
Fair																		
Not So Good	•			•		•		÷	•	•				•	•	•	•	5
Poor	•	•	ł	•	•		ł	•	•	•	•	•	•	•	•	•	·	6

CLASSIFICATION

Ask All Respondents:

Now I just need to check on a few points to help us classify our data.

- 15(c) For how long have you lived in the Metropolitan area? [Circle below in column 15(a).]
- 15(c) And for how long have you lived at your present address? (Circle below in column 16(b).]

	Q.15(a) Metropolitan Area	Q.15(b) Present Address
Less than 1 year	. 1	1
1-2 years	. 2	2
3-4 years		3
5-6 years		4
7-10 years		5
More than 10 years	. 6	6

16(a) In total, how many cars are owned by the members of your household?

1	
2 [ASK Q.16(b)] 2	
3	
More than 3 4	
None (SKIP to Q.17)	
Refused	

16(b) To what extent do you normally have personal use of a car available to you? (READ LIST.)

Always 1	
Most of the time	
Part of the time	
Occasionally 4	
Never	

17. What was the last grade you completed in school?

Eighth grade or less 1	
Some high school	
Completed high school	
Some college	
Completed college 5	
Graduate school	
Refused	

18(a) What is your occupation/the occupation of the male head of the household? (Probe for specific job. If unemployed, ascertain whether respondent/male head is retired, disabled or temporarily unemployed.)

(Ask if respondent is female head or if household has no male head and respondent is not female head:)

18(b) What is your occupation/the occupation of the female head of the household? (Probe for specific job. If unemployed, ascertain whether respondent/female head is retired, disabled or temporarily unemployed.)

(Ask if respondent is neither male head nor female head:)

18(c) What is your occupation? (Probe for specific job. If unemployed, ascertain whether retired, disabled or temporarily unemployed.)

[If respondent is student, ask Q.18(d). Otherwise, go to end.]

18(d) Do you work either full time or part time while you are going to school?

Yes [ASK Q.18(e)] 1 No (GO TO END) 2

18(e) What do you do at work? (Probe for specific job.)

THANK YOU VERY MUCH FOR YOUR COOPERATION!

Record respondent data in box on page 1. Make sure completed referral sheet is attached to com pleted questionnaire!

Validated By:_____ Date:_____

(ATTACH TO COMPLETED INTERVIEW)

REFERRAL SHEET

Name	Age	Work	H.S. Stud	Tech/ Coll. Stud	Have Lic	Make Trips	At Home Now	Respondent Interviewed
Male Head/Household								
		1	1	1	1	1	1	1
Other Males								
		2	2	2	2	2	2	2
		3	З	3	3	3	3	3
		4	4	4	4	4	4	4
		5	5	5	5	5	5	5
Female Head/Household								
		6	6	6	6	6	6	6
Other Females								
		7	7	7	7	7	7	7
		8	8	8	8	8	8	8
		9	9	9	9	9	9	9
		0	0	0	0	0	0	0
None								TERMINATE

Of last 10 work trips, number of times rode bus: (WRITE IN:)

Likelihood of using bus for *work* trips (CIRCLE ONE:)

Extremely												
Very						•						1
Quite												
Rather												
Not too		•	•		•							2
Not at all												

Of last 10 shopping trips, number of times rode bus: (WRITE IN:)

Likelihood of using bus for *shopping* trips (CIRCLE ONE:)

Extremely												
Very					•	•		•				1
Quite												
Rather												
Not too												2
Not at all												

EXHIBIT C Examples of Secondary Information Sources

1. U.S. Census

Socioeconomic, demographic and housing data are available in census tracts or block groups, always aggregated, sometimes incomplete.

١

Critical Information

- 1. Location of place of work
- 2. Income distribution
- 3. Automobile availability
- 4. Mode to work
- 5. Age distribution

Available on tape or hard copy, usually from planning agency.

2. Planning Agencies

Current Land Use Plan Current Zoning

Raw Data

- 1. Employment locations
- 2. Centers of Commercial Activity sales tax data

Locations of Special Trip Generators

- 1. Elderly Housing
- 2. Medical Facilities
- 3. Schools
- 4. Employment
- 5. Shopping

3. Traffic Department

Traffic Volumes/Capacity Intersection Geometry/Control Parking and Other Regulations Construction Plans

4. In-House Data Collection

Running Time Information Time Between Check Points Schedule Adherence Running Time on Alternative Routes

Passenger Comments/Complaints Driver Comments/Complaints

Ridership Counts Rate Revenue Off-On Counts # of Transfers Issued/Accepted

5. Community Sources

Political Leaders

Other Community Leaders/Knowledgeable Persons Informal Sources Task Forces

Public Hearings

6. Private Sources

Utility Records and Surveys Newspaper Surveys and Market Information Chamber of Commerce Surveys, Local Area Promotional Information

EXHIBIT D

Comparative Assessment of Alternative Survey Methods

Criteria Method	Average Costs	Response Rate	Data Recovery Time	Rapport	Call Back/ Follow-up	Normal Length		
Telephone	moderate	highest	immediate	moderate	easy low cost, quick	10 to 30 minutes		
Personal Interview	highest	highest	moderate/ immediate	highest	difficult, costly, slow	15 to 45 minutes		
On-Board	lowest	lowest	moderate/ immediate	lowest	easy, moderate cost, very slow	10 to 20 minutes		
Advantage	On-Board	Personal Interview	Telephone	Personal Interview	Telephone/ On-Board	On-Board		

(from Guide to Market Research in Public Transit)

× . •

23488

HE 4451 .T83 1988

Transit marketing management market research

DATE DUE

MTA LIBRARY

ONE GATEWAY PLAZA, 15th Floor LOS ANGELES, CA 90012

